Author's response to reviews

**Title:** Prospective Blind Comparative clinical study of two point fixation of zygomatic complex fracture using wire and mini plates.

**Authors:**

Lakshmi N Gandi (lakshmigandi@yahoo.co.in)
Vivekanand S Kattimani (drvivekanandsk@gmail.com)
Sridhar S Meka (mfmsr@ymail.com)
Srinivas P Chakravarthi (drsrinivaschakravarthi@gmail.com)
Amit V Gupta (dramitvgupta@gmail.com)

**Version:** 3 **Date:** 10 January 2012

**Author's response to reviews:** see over
Covering letter

Dear Sir,

Here by I am submitting original work titled “Prospective Blind Comparative clinical study of two point fixation of zygomatic complex fracture using wire and mini plates.” to your esteemed journal for publishing on behalf of all authors.

The unique multi-center study using classification mentioned by Spissel and Schroll operated by team of oral and maxillofacial surgeons and assessed by maxillofacial surgeons during follow up visits to assess the effectiveness of both fixation systems in ZMC fracture and observer is blinded to the procedure carried out.

The revision done according to journal format with relevant changes necessary for each particular section.

Thanking you,

Dr.Vivekanand.S.Kattimani MDS
Asst.Prof.
Department of Oral and Maxillofacial Surgery,
Saraswati Dhanwantari Dental College and Hospital, Parbhani.
Maharashtra, India.431401.
Email:drvivekanandsk@gmail.com.
Mb:-00-91-9689742418.